\begin{table}%T1
%\centering
\small
\newcommand{\nop}   {\multicolumn{2}{c}{\ }}
\newcommand{\nt}    {{\small ${\dag}$}}
\caption{\label{tbl-published} 
  Observational data of the PMS stars. The spectral classifications based on 
  different criteria are separated by (/) to distinguish them from the given 
  spectral ranges ($-$).
  Additional information on the double stars (denoted by \ddag\ in Col.~2) is 
  given in Table~\ref{tbl-binsys}. Uncertain values are indicated with (:).
  The foreground colour-excess \Eis\ is labelled with (*) if \Ris~$\neq$~3.1 
  (Appendix~\ref{sec-for-ext}).
  The $h$ in Col.~7 denotes Hipparcos values (ESA 1997),
  see Appendix~\ref{sec-app-b} for supplementary information on each star.
  Column~9 indicates if Walraven photometry was obtained for the star.
  The inclination is indicated with E or P based on the methods (X, Y, Z)
  described in Appendices~\ref{sec-inca} and~\ref{sec-app-c}.
} %% end-caption
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\par
\renewcommand{\rmi}     {{\footnotesize{\rm I}}}
\renewcommand{\rmii}    {{\footnotesize{\rm II}}}
\renewcommand{\rmiii}   {{\footnotesize{\rm III}}}
\renewcommand{\rmiv}    {{\footnotesize{\rm IV}}}
\renewcommand{\rmv}     {{\footnotesize{\rm V}}}
\newcommand{\rmx}       {{\footnotesize{\rm III-II}}}
\newcommand{\rmy}       {{\footnotesize{\rm III-IV}}}
\newcommand{\rmz}       {{\footnotesize{\rm I-II}}}
\renewcommand{\upc}     {\multicolumn{2}{c}{(pc)}}
\newcommand{\cstar}     {\multicolumn{3}{c}{PMS star}}
\newcommand{\cVphot}    {\multicolumn{3}{c}{$V_{\rm phot}$}}
\newcommand{\cdis}      {\multicolumn{2}{c}{$d$}}
\newcommand{\cspec}     {\multicolumn{2}{c}{Spectral classification}}
\newcommand{\cvsini}    {\multicolumn{2}{c}{$v \sin i$}}
\newcommand{\magnitude} {\multicolumn{4}{c}{(magnitude)}}
\newcommand{\TCD}       {\multicolumn{1}{c}{TCD}}
\newcommand{\VIS}       {\multicolumn{1}{c}{visual spectra}}
\newcommand{\ukms}      {\multicolumn{2}{c}{(\kms)}}
\newcommand{\hp}        {$^{h}$} %%% \hbar
\newcommand{\hq}        {$^{h:}$} 
\newcommand{\ltv}       {{$<$}5}
%%%         \BD}[2]     {${\rm BD{#1}}^{\circ}{#2}$}
\newcommand{\BFOri}     {{\BD{-06}{1259}}}
\newcommand{\FUOri}     {{\BD{+09}{5427}}}    %%  {{Nova Ori 1937}}
\newcommand{\RCrA }     {{\CD{-37}{13027}}}
\newcommand{\NXPup}     {{\CD{-44}{3318}}}
\newcommand{\VOri }     {{\BD{-06}{1253}}}
\newcommand{\NVOri}     {{\BD{-05}{1324}}}
\newcommand{\BDX}       {{\BD{-08}{1128}}} %% HD~35929
\newcommand{\BDXX}      {{\BD{-17}{6127}}} %% HD~199143
\newcommand{\BDXXX}     {{\BD{+26}{887a}}} %% RR Tau    <> HD~245906 = BD+26 887
\newcommand{\CDX}       {{\CD{-42}{10650}}}%% HD~139614
\newcommand{\CDXX}      {{\CD{-27}{10778}}}%% HD~144432
\newcommand{\CDY}       {{\CD{-52}{4340}}} %% HD~98922
\newcommand{\CDYY}      {{\CD{-70}{775}}}  %% HD~95881 
\newcommand{\appone}    {Appendix~\ref{stars-104237} }
\newcommand{\apptwo}    {Appendix~\ref{stars-98922} }
\newcommand{\apptre}    {Appendix~\ref{stars-95881} }
\newcommand{\mcx}       {\multicolumn{3}{r}{X}}
\newcommand{\nix}  {\multicolumn{2}{c}{~~~}}
\newcommand{\nostar}    {--}
\begin{tabular}{  r l@{}c@{~}l   r@{}c@{}l l@{}c l l   r@{}l r@{}l  c   c@{~} r@{} l@{~} c@{\ } l }
\kline \hline 
\mycl{1}{(1)}&\mycl{3}{(2)}&\mycl{3}{(3)}&\mycl{2}{(4)}&\mycl{1}{(5)}&\mycl{1}{(6)}&\mycl{2}{(7)}&\mycl{2}{(8)}&\mycl{1}{(9)}&\mycl{2}{(10)} & & & \\
\#&        \cstar  		 &\cVphot         &\Eis& &    \cspec               	&\cdis   &     \cvsini  &w   &$i$ & & & &              \\
\cline{5-8} \cline{10-11} \cline{17-21}\\[-3.2mm]
  &   	 &      &		 & \magnitude     & &\TCD & \VIS            &\upc         &\ukms	        &   &\mcx    &Y  &Z            \\
\kline\\[-3.2mm]
 1&XY Per   &\ddag &HD~275877 & 9   &--&10.5  &0.19& &A1-A3 &A2\rmii e        & 350&$^{15}$ &130 &:+95:$^{13},217^{44}$ &--&E&r&: &12 &\rmii B-R$^{25}$\\
 2&WW Vul   &      &HD~344361 &10.4 &--&12.6  &0.3:& &A2-A5 &A0-3\rmv e       & 534&$^{16}$ &125 &$^{40},220^{44},150^{45}$ 
 &+&E&r&  &12 &\rmii BR$^{26}$\\
 3&UX Ori   &\ddag &HD~293782 & 8.7 &--&12.8  &0.06& &A2-A3 &A3\rmiii e       & 400&        &~70 &$^1,215^{44},140^{45}$&+&E&r&  &18 &\rmiii Rm$^{26}$\\
 4&HD~163296 &      &MWC~275  & 6.79&--&7.01  &0.00& &B9-A0 &A0/A7\rmv e      & 122&\hp     &120 &$^{2,3},133^{44}$     &+&E&n&  &24 &\rmx B$^{17}$\\
 5&HD~150193 &\ddag &MWC~863  & 8   &.&83    &0.03& &A3-A5 &A4\rmv e         & 150&\hp     &100 &$^2$                  &+&E&n&  &21 &\rmiv B$^{17}$\\
 6&\fnv     &\ddag &HD~283817 &10.30&--&10.56 &0.04& &F-G   &A3e              & 150&$^{41}$ &-- &                      &+&E&o&: &17 &\rmii B$^{20}$\\
 7&BF Ori   &\ddag &\BFOri   & 9.7 &--&13.5  &0.06& &A3-A5 &A5\rmii-\rmiii e & 450&        &100 &$^1,80^3,37^{44},40^{45}$
                                                       &+&E&r&  &20 &\rmiii Rm$^{26}$ \rmii B$^{17}$\\
 8&HD~31648 &      &MWC~480  & 7   &.&7     &0.02& &A5-A6 &A2-3\rmv e       & 131&\hp&~80 &$^{34},102^{44}$           &+&E&n&  &24 &\rmiii$^{19}$\\
 9&HR~5999  &\ddag &V856~Sco & 6.72&--& 8.2  &0.03& &A5-A7 &A5-7e\rmiii-\rmv & 210&\hp&180 &$^{14}$          &+&E&r&  &17 &\rmii B$^{14,21,26}$\\
10&\BPC	    &      &HD~39060 & 3   &.&85    &0.00& &A5-A7 &A5\rmv-\rmiv     & 19.3&\hp&120&$^4$             &--&E&n&     &12 &--       \\
11&HD~104237 &      &DX~Cha   & 6   &.&60    &0.02& &A5-A7 &A4\rmv e         & 116&\hp&150 &$^5$             &+&E&n&     &26 &\rmy B$^{19}$\\
12&HD~169142 &\ddag:&MWC~925  & 8   &.&11    &0.00& &B9    &A5\rmv e         & 145&$^{30}$ &~55 &$^{31}$     &--&P&n&     & 7 &\rmi$^{32}$\\
13&HD~36112 &      &MWC~758  & 8   &.&26    &0.06& &A5-A7 &A3e:             & 200&\hp&~60 &$^{33}$          &--&E&n&     &20:&\rmiv B$^{22}$\\
14&HD~139614 &\ddag:&\CDX     & 8   &.&26    &0.02& &A7    &A7\rmv e         & 157&$^{30}$ &~13&$^{29}$,24$^{31}$
                                                                                                    &--&P&n&     & 5 &\rmi$^{32}$\\
15&HD~142666 &      &V1026~Sco& 8.65&--& 8.84 &0.22& &F2\rmv&A3/A8\rmv e      & 114&$^{30}$ &~56&$^{29},70^{31},72^{44}$ 
                             &--&E&r&:    &17 &\rmii R$^{32}$\\
16&V346 Ori &      &HD~287841 &10.0 &--&11.9  &0.06& &A3-A4 &A5\rmiii e       & 450&   &-- &                 &+&E&r&     &16 &\rmi$^{19}$\\
17&V350 Ori &      &\nostar  &10.3 &--&13.2  &0.06& &A4-F0 &A0-A1\rmiv e     & 450&   &-- &                 &+&E&r&     &17 &\rmii BR$^{13,19}$\\
18&V351 Ori &      &HD~38238 & 8.3 &--&11.6  &0.06& &A3-A7 &A7\rmiii e       & 450&   &-- &                 &+&E&r&\nt  &20 &\rmiii R$^{19}$\\
19&BN Ori   &      &HD~245465 & 9   &.&65    &0.07& &A5-F2 &A7-G9\rmiii-\rmiv e& 450&   &220 &$^6$             &+&E&r&\nt  &17 &\rmii BR$^{6,25}$\\
20&HD~35929 &      &\BDX     & 8   &.&13    &0.06& &A5-A7 &A8\rmiii :       & 450&   &150 &:$^{12}$         &--&E&r&:    &15 &\rmi$^{19}$\\
21&HD~144432 &      &\CDXX    & 8   &.&15    &0.00& &A7\rmv&A9\rmv e         & 200&\hp&~54 &$^{29},74^{31},85^{44}$
 &--&E&n&:    &15 &\rmy B$^{27,32}$\\
22&\fnz     &      &SAO53184 & 9   &.&30    &0.05& &--&A9\rmiii           & 260&\hp&-- &                 &--&--&n&     &13 &--              \\
23&FU Ori   &      &\FUOri   & 9   &.&3     &0.06& &G0-G2 &F2/G\rmi-\rmii e/K0& 450&   &~60 &$^8$             &+&E&r&\nt  &40 &\rmiv B$^{26}$\\
24& R CrA   &      &\RCrA    &10.0 &--&14.4  &0.18&*&F2pe  &A5-F5pe            & 140&\hq&-- &                 &+&E&r&     &14 &\rmii RB$^{26,28}$\\
25&CQ Tau   &      &HD~36910 & 9.9 &--&11.6  &0.07& &A8    &A8-F2\rmiii e      & 100&\hp&110 &$^1,105^{44},90^{45}$  &+&E&r& &40 &\rmy R$^{23}$\\
26&NX Pup   &\ddag &\NXPup   & 9.8 &--&11.2  &0.00& &F0-F4 &B9-F0-2\rmiii e    & 500&   &120 &$^1$             &+&P&o&     &12 &\rmii B$^{26}$\\
27&\fnp     &\ddag &V1578~Cyg&10.1 &--&10.2  &0.31& &A5    &A2-A4\rmiii e      & 900&$^{39}$ &150 &$^1,100^3$  &--&E&--&     &19 &\rmiv B$^{17}$\\
28&HK Ori   &\ddag &MWC~497  &11   &.&6     &0.06& &F5    &B8/A4e             & 450&   &175 &$^3$,150$^{40}$  &+&P&o&     &10 &\rmz B$^{26}$\\
29&AK Sco   &\ddag &HD~152404 &11.0 &--&11.4  &0.13& &F5-F6 &F5\rmiv e          & 150&\hp&~19 &$^9$             &+&E&r&     &20 &\rmii R$^{26}$\\
30& Z CMa   &\ddag &HD~53179 & 8.8 &--&11.2  &0.16&*&F-G   &B5-8/F5-7\rmi e    &1050&$^{37}$ &100&$^{38}$      &+&E&o&     &40 &\rmiv B$^{26}$\\ 
%%%%%%%	                                         Ibragimov \& Shevchenko \cite{Ibragimov90};  930$\pm$120pc  
31&HD~199143 &\ddag &\BDXX    & 7   &.&1     &0.0 & &--&F8\rmv e           & ~48&\hp&200 &:$^{36},155^{44}$ &--&--&--&     &13 &em$^{36}$\\
32&V380 Ori &\ddag &\VOri    &10.0 &--&11.5  &0.06& &F     &B8-A2e             & 450&   &~20 &$^3$,200$^{40}$  &+&P&o&     &22 &\rmi$^{26}$\\
33&NV Ori   &      &\NVOri   &10.0 &--&10.6  &0.06& &F0-F2e&F2-8\rmiii-\rmiv e & 450&   &~80 &$^1,81^{44}$     &+&E&r&     &20 &em$^{13}$\\
34&RR Tau   &      &\BDXXX   &10.2 &--&14.2  &0.07& &A4-F0 &B8-A2\rmii-\rmiii e& 600&$^{24}$ &$>$60&$^3,225^{44},140^{45}$  &--&E&r& &20 
&\rmii R$^{17}$\\
35&KK Oph   &\ddag &\nostar  & 9.4 &--&12.8  &0.32& &A5    &A5-7\rmv e       &$<$250&$^{35}$ &~85 &$^{3,12},177^{44}$ &+&E&r&  &22 &\rmii R$^{19}$\\ 
36&SU Aur   &      &HD~282624 & 9.0 &--&11.8  &0.12& &F5-G2 &G2\rmiii e/K:      & 144&\hp&~66 &$^{10}$          &--&E&r&     &40 &\rmii B$^{25}$\\
37&RY Tau   &\ddag &HD~283571 &10.42&--&11.12 &0.04& &F8-K1 &K1\rmiv-\rmv e/K5: & 134&\hp&55  &$^{44}$          &--&E&r&     &35 &\rmii B$^{18}$\\
%%%T Tau    &\ddag &HD~284419 & 9.3 &--&13.5  &0.04& &F8-K1 &K0\rmiv-\rmv e/K5: & 160&   &~19 &.5$^{10}$        &--&P&r&     &20 &\rmiii B$^{26}$\\
38&HD~283572 &      &V987 Tau & 8.9 &--& 9.1  &0.03& &G2-G5 &G5\rmiv e          & 128&\hp&130 &$^{11}$,80$^7$   &--&E&o&     &40 &abs            \\
\kline\\[-3.2mm] 
39&\fnn     &      &HD~245185 & 9.83&--& 9.9  &0.06& &B9    &B9/A5\rmv e        & 450&   &~80 &$^2$,150$^{40}$  &--&P&n&     &\ltv&\rmi$^{25}$\\
40&HD~100546 &      &KR~Mus   & 6.67&--& 6.77 &0.00& &A0    &B9\rmv e           & 103&\hp&250 &$^{12}$          &+&E&n&     &\ltv&\rmii R$^{12}$\\
41&HD~98922 &\ddag &\CDY     & 6.7~&--& 6.8~ &0.05& &A0    &B9\rmv e           & 200&$^{42}$ &-- &            &+&E&--&     & 6 &\rmiv B$^{19}$\\
42&HD~97048 &      &CU~Cha   & 8.38&--& 8.56 &0.06&*&B9-A0 &B9.5-A0e           & 180&\hp&140 &$^{1,2}$         &+&E&n&     & 8 &\rmiv B$^{26}$\\
43&HD~95881 &      &\CDYY    & 8.2~&--& 8.4~ &0.17& &--&A1\rmiii e         & 300&$^{43}$ &150 &$^{12}$     &--&E&--&     &11 &\rmii B$^{19}$\\
\kline \end{tabular}
%%%%%%%%%%%%%%%%%%%%%%%%
\par
\medskip
\hspace*{-5mm}\begin{tiny}\begin{tabular}{r@{:~}l r@{:~}l r@{:~}l r@{:~}l r@{:~}l}
1& B\"{o}hm \& Catala (\cite{Bhm95}) & 10& Bertout et~al. (\cite{Bertout88}) &  19& \dW\ et~al. (\cite{DW01}) & 28& Hamann \& Persson (\cite{Hamann92}) & 37& Shevchenko et~al. (\cite{Shevchenko99})   \\
2& Finkenzeller (\cite{Finkenzeller85}) &  11& Walter et~al. (\cite{Walter87})         &  20& Martin (\cite{Martin93})               &  29& Meeus et~al. (\cite{Meeus98})        &  38& Hartmann et~al. (\cite{Hartmann89})     \\
3& Hillenbrand et~al. (\cite{Hillenbrand92}) &   12&  Grady et~al. (\cite{Grady96})          &  21& Bessel \& Eggen (\cite{Bessell72})      &30& Sylvester et~al. (\cite{Sylvester96})    &39& Elias (\cite{Elias78b})              \\
4& Slettebak (\cite{Slettebak82})          & 13& Herbig \& Bell (\cite{Herbig88})        & 22&  Pogodin (\cite{Pogodin95})              & 31& Dunkin et~al. (\cite{Dunkin97a})      &40& Corporan \& Lagrange (\cite{Corporan99})\\
5&  \appone                 &14& \TAD\ et~al. (\cite{Tad89})          &23&  Berdyugin et~al. (\cite{Berdyugin90})     &32& Dunkin et~al. (\cite{Dunkin97b})      &41&  Weaver \& Hobson (\cite{Weaver88})   \\
6&  Shevchenko et~al. (\cite{Shevchenko97})  &15& Herbertz et~al. (\cite{Herbertz91})       &24& Artyukhina (\cite{Artiyukhina60})           &33& Beskrovnaya et~al. (\cite{Beskrovnaya99})  &42& \apptwo                  \\
7& Joncour et~al. (\cite{Joncour94})     &16&  Friedemann et~al. (\cite{Friedemann93})     &25& Fern\'{a}ndez et~al. (\cite{Fern95}) &34&  Glagolevskii et~al. (\cite{Glagolevskii98}) &43&  \apptre                  \\
8&   Herbig (\cite{Herbig66})             &17&   Finkenzeller \& Mundt (\cite{Finkenzeller84}) &26&   Reipurth et~al. (\cite{Reipurth96})      &35&   Wallenquist (\cite{Wallenquist37})         &44&  Mora et~al. (\cite{Mora01})         \\
9&   Andersen et~al. (\cite{Andersen89})   &18&   Kolotilov \& Zajtseva (\cite{Kolotilov74}) &27&  \Perez\ et~al. (\cite{Perez04})       &36&  v.d.Ancker et~al. (\cite{vdA00})   &45&  Grinin et~al. (\cite{Grinin01})       \\
\end{tabular}\end{tiny}
%\vspace*{0.8mm}
\end{table}